Author's response to reviews

Title: Unique Challenges for Appropriate Management of a Teen with Superior Mesenteric Artery Syndrome as a Result of Anorexia Nervosa: A Case Report

Authors:

Philip A Verhoef (pverhoef@mednet.ucla.edu)
Angelika Rampal (arampal@mednet.ucla.edu)

Version: 2 Date: 20 March 2008

Author's response to reviews:

We revised the manuscript to reflect the formatting for the abstract, to correct the formatting for the title page, and we added an additional conclusions section to the discussion.